Author's response to reviews

Title: A study of head and neck cancer treatment and survival among Indigenous and non-Indigenous people in Queensland, Australia, 1998 to 2004

Authors:

Suzanne P Moore (moores@fellows.iarc.fr)
Adele C Green (adele.green@qimr.edu.au)
Gail Garvey (gail.garvey@menzies.edu.au)
Michael D Coory (michael.coory@mcri.edu.au)
Patricia C Valery (Patricia.Valery@menzies.edu.au)

Version: 9 Date: 24 October 2011

Author's response to reviews: see over
Monday, 24 October 2011

Mr. Danrolf de Jesus  
Tel: +44 (0) 20 3192 2013  
e-mail: editorial@biomedcentral.com

Dear Sir,

Re: Resubmission of a revised manuscript 9278027885481378

Thank you accepting our manuscript entitled ‘A study of head and neck cancer treatment and survival among Indigenous and non-Indigenous Australians in Queensland Australia, 1998 to 2004’ for publication in BMC Cancer. We have made one additional formatting change and attach an explanation below.

We would like to re-confirm that this manuscript has not been submitted or published elsewhere and that no conflict of interest exists. All authors have contributed to and agree with the final revised version.

Thank you again for accepting our manuscript for publication in BMC Cancer.

Yours sincerely,

Suzanne P Moore

Formatting changes:

Minor revisions:

Suzanne Moore has amended her email address both in the manuscript and in the Biomedical Central authors list to: moores@fellows.iarc.fr